Challenges and Methodologies of Fully Automatic Whole Heart Segmentation: A Review

Xiahai Zhuang*

Shanghai Advance Research Institute, Chinese Academy of Sciences, China

Submitted November 2012. Accepted for publication March 2013.

ABSTRACT

Whole heart segmentation from magnetic resonance imaging or computed tomography is a prerequisite for many clinical applications. Since manual delineation can be tedious and subject to bias, automating such segmentation becomes increasingly popular in the image computing field. However, fully automatic whole heart segmentation is challenging and only limited studies were reported in the literature. This article reviews the existing techniques and analyzes the challenges and methodologies. The techniques are classified in terms of the types of the prior models and the algorithms used to fit the model to unseen images. The prior models include the atlases and the deformable models, and the fitting algorithms are further decomposed into four key techniques including localization of the whole heart, initialization of substructures, refinement of boundary delineation, and regularization of shapes. Finally, the validation issues, challenges, and future directions are discussed.

Keywords: whole heart segmentation, cardiac segmentation, cardiac atlas, segmentation propagation, statistical shape model, deformable model

1. BACKGROUND AND MOTIVATION

According to World Health Organization (WHO), cardiac vascular diseases are the number one cause of deaths globally. An estimated 17.3 million people died from cardiac vascular diseases in 2008, accounting for 30% of deaths around the world [1]. To improve diagnosis and treatment of cardiac vascular diseases, there has been much emphasis on developing and introducing novel technologies in medical imaging and image computing to clinical practice and medical research of cardiology [2–5]. The morphological and pathological information from the three-dimensional (3D) medical imaging has made revolutionary impacts on cardiology, including the cardiac computed tomography (CT) or computed tomography angiography (CTA) [6], and the cardiac magnetic resonance imaging (MRI) [7]. The 3D data, with good contrast and wide fields of view, provide valuable anatomical information of the heart. To enable the development of new clinical applications and thus improving cardiology, accurate

^{*}Corresponding author: Xiahai Zhuang, Shanghai Advance Research Institute, Chinese Academy of Sciences, No. 3 Building, 99 Hai Ke Road, Shanghai 201210, China. Phone: (8621) 2030-5896. Fax: (8621) 2030-5889. E-mail: <u>zhuangxh@sari.ac.cn</u>.

extraction and precise interpretation of this anatomical information becomes particularly important.

The goal of whole heart segmentation is to extract the volume and shapes of all the substructures of the heart, commonly including the blood cavities of the four chambers, left ventricle myocardium, and sometimes the great vessels as well if they are of interest [8–24]. This is in contrast with many other cardiac segmentation works in the literature, which focus solely on the ventricles and/or left ventricle myocardium [25–44], or isolation of the whole heart volume [45–47]. Figure 1 shows a whole heart segmentation result from a cardiac CTA volume. The blood cavities of the four chambers are:

- (1) The left ventricle blood cavity (referred to as *left ventricle* or short as LV).
- (2) The right ventricle blood cavity (referred to as *right ventricle* or short as **RV**).
- (3) The left atrium blood cavity (referred to as *left atrium* or short as LA).
- (4) The right atrium blood cavity (referred to as *right atrium* or short as **RA**).



Figure 1. Whole heart segmentation result from a cardiac CTA volume. (a) A slice from the axial view of a cardiac CTA volume. (b) The corresponding whole heart segmentation labels, in different color. (c) The whole heart segmentation labels superimposed onto the CTA image. (d) The volume rendering of the whole heart segmentation labels.

The myocardium of the left ventricle is also of interest in the whole heart segmentation:

(5) The phrase *myocardium* (MYO) is commonly used to refer to that of the left ventricle, and *endocardium* (ENDO) and *epicardium* (EPI) are used to refer to the endocardial surface and epicardial surface of the left ventricle, respectively.

The great vessels include the aorta and the pulmonary artery. Due to the different fields of view, the coverage of the great vessels can be different in different scans. To have a consistent definition of the great vessels across different subjects, the major trunks of the vessels can be defined as follows:

- (6) The aorta is defined as the beginning trunk of the ascending aorta from the aortic valve to the superior level of the atria (referred to as *ascending aorta* or AO); for example, the mean length of this trunk is about 41.8 mm as we measured in 25 healthy subjects.
- (7) The pulmonary artery is defined as the beginning trunk between the pulmonary valve and the bifurcation point (referred to as *pulmonary artery* or **PA**); for example, the mean length of this trunk is about 41.9 mm as we measured in 25 healthy subjects.

Other (parts of) great vessels were also segmented, including the descending aorta, the pulmonary veins, the superior and inferior vena cava, and the coronary sinus [8].

Whole heart segmentation results are important to clinical studies and applications, including the following:

- To extract functional indices such as chamber volumes, ejection fraction (EF), myocardial mass, myocardial thickness/thickening/motion, etc. These functional indices are important measures in clinics [4, 48–54]. For example, the EF determines how well the heart is pumping out blood and is important in diagnosing and tracking heart failure [51, 52]; the myocardial motion is linked to the contractility of the heart and is shown to be strongly correlated to the remodeling of post cardiac resynchronization therapy [53, 54]. It is also anticipated that the functional analysis of the whole heart may detect subtle functional abnormalities or changes of the heart, and this is important to patients who otherwise have normal systole in ventricles but are suspected to have abnormal functions in other regions [55, 56].
- To enable 3D surface rendering, as shown in Figure 1 (d). The rendering result has a wide range of applications such as investigating the congenital cardiac malformations in a morphological study [57]. Also, the latent definition of the landmarks from the whole heart segmentation result can define the short axis (SAX) and long axis (LAX) to display any desired two-dimensional (2D) views, such as the four-chamber view and SAX view [58]. The geometrical information from the whole heart segmentation is useful in guiding interventional procedures by fusing the 3D surface of the heart constructed from an anatomical scan to the real-time interventional scans [59]. For example, in the radio frequency ablation surgery, the surgeons can obtain a better relative position of the interventional catheter tip to the heart by setting certain transparence to the 3D surface model. Also, the landmarks from the segmentation result enable fast navigation of the catheter tip during surgeries, such as accessing the pulmonary veins in the radio frequency ablation surgery.

The goal of this work is to review the whole heart segmentation studies in the literature, and discuss the challenges and methodologies for fully automating such segmentation. It should be noted that this review does not cover articles and methods for ventricle segmentation or isolation of the heart, topics already reviewed in previous papers [47, 60–66]. The rest of this paper is organized as follows. Section 2 presents the major challenges of the fully automatic whole heart segmentation. Section 3 reviews and analyzes the methodologies of the existing works. The databases used to search the papers include the PubMed (http://www.ncbi.nlm.nih.gov/pubmed), the IEEE Xplore (http://ieeexplore.ieee.org/Xplore), Google Scholar (http://scholar.google.com/), and all the papers published in IEEE Transaction on Medical Imaging and Medical Image Analysis since 2002. Section 4 discusses the validation issues. Section 5 discusses the remaining challenges and potential solutions. Section 6 concludes this work.

2. CHALLENGES

Manually delineating all the substructures of the whole heart is labor-intensive and tedious. More importantly, manual delineation is subject to intra- and inter-observer variations. Therefore, automatic whole heart segmentation has become increasingly popular in recent years. This fully automated processing is particularly useful in the clinical studies where a large number of images need segmentation. However, fully automatic whole heart segmentation is difficult. The challenges are the following.

Firstly, the heart organ, with multiple chambers and great vessels, is complex in geometry. The heart shape can vary significantly in different subjects or from the same subject at different cardiac conditions. This shape variation is especially evident when pathological cases are involved. For example, Figure 2 (a) shows an MRI volume from a healthy volunteer and (b) is from a patient with right ventricle *hypertrophy* (an abnormal, severely dilated right ventricle). The segmentation using a prior model constructed from a training set of healthy volunteer data performed well on the healthy subject, as shown in Figure 2 (a). However, the same segmentation tool achieved much poorer result when it was used to segment the pathological data, as shown in Figure 2 (b). The shape variation represents the major challenge for the automated whole heart



Figure 2. (a) MRI volume from a healthy volunteer with successful segmentation, defined by the green contour. (b) MRI volume from a patient with right ventricle hypertrophy, with erroneous segmentation using a traditional atlas propagation and segmentation scheme, defined by the red contour. Reproduced with permission [23].

segmentation, even when the statistical shape priors are used to estimate the *unseen* cases. *Unseen* refers to the medical images that need segmentation. It is practically difficult, if not impossible, to capture all possible heart shapes from different pathologies using a prior model trained by a limited training dataset.

Secondly, some boundaries between anatomical substructures are visually indistinct based on the *intensity distributions*, namely texture patterns, of the medical images. This happens in situations including the following:

- (1) Some substructures are not separated by physical tissues or structures; instead their boundaries are defined based on anatomical knowledge. These boundaries include the valve planes separating the atria and great vessels from the ventricles, the boundaries between the left atrium and pulmonary veins, and the boundaries between the right atrium and superior/ inferior vena cava. Figure 3 (a) shows an example of a cardiac MRI image, where the tricuspid valve plane was delineated at a wrong location by a traditional atlas propagation and segmentation scheme [29].
- (2) The intensity distributions between adjacent tissues or substructures are similar or even identical. For example, the intensity levels of the papillary muscle and myocardium are similar while the papillary muscle tends to be included in the ventricular cavities. The intensity of the myocardium is also similar to the adjacent liver and body muscle in cardiac CT data. In cardiac CTA, the intensity distribution of the unenhanced cavities, such as the right ventricle, is also similar to that of the myocardium. The cardiac CTA image in Figure 3 (b) shows that the intensity distributions of the myocardium, the right ventricle, and the adjacent tissues of the body muscle and liver are similar. It is therefore difficult to separate these regions from each other if a segmentation method does not utilize prior models for guidance and solely relies on the image intensity. Figure 3(c) shows an example of incorrect segmentation of the myocardium by a traditional atlas propagation and segmentation method, due to the indistinct boundary [29].
- (3) The thin walls of the atria and vessels, as well as the partial volume effects, make the boundaries between these substructures unclear when they are adjacent to each other. As illustrated in Figure 3 (d), an axial view of a cardiac MRI volume shows that the boundaries between the left atrium, the right atrium, and the aorta are unclear.

Due to these indistinct boundaries, the *lower level* techniques, which do not use shape priors for guidance [62–65], have difficulty achieving the fully automatic whole heart segmentation.

Finally, due to the complex motions and blood flow within the heart, the imaged data may contain heavy motion artifacts, intensity inhomogeneity, and noise. This contributes to the unsmooth and suboptimal delineation of the boundaries. Smoothness constraints and shape regularization are generally adopted to deal with this challenge.

2.1. Key Techniques for Automatic Whole Heart Segmentation

Automatic segmentation of complex organs, such as the heart, is generally formulated as a fitting process from a prior model to the unseen images for segmentation propagation [9–22]. The prior models contain the segmentation information and are used to guide the segmentation procedure, and the fitting procedure is generally implemented in a *hierarchy* manner.



Figure 3. Indistinct boundaries. (a) A cardiac MRI image with segmentation result. In the MRI image, no flat tricuspid valve plane between the right ventricle and the atrium is visible, and the segmentation incorrectly delineates the plane, as indicated by the red arrow. However, the true boundary should be at the white dash line position. (b) Cardiac CTA image showing that within the elliptic region (marked by orange boundary), there are three tissues or substructures with similar intensity distributions, including the myocardium (septum), the right ventricle, and the liver tissue. The intensity distributions between the myocardium and the body tissue adjacent to the heart, as the arrow indicates, are also similar. (c) An example of erroneous segmentation on myocardium which has a similar intensity distribution as the adjacent liver tissue. (d) Cardiac MRI image showing that the boundaries between the left atrium, right atrium, and the aorta are unclear, as indicated by the arrow, due to the thin walls and partial volume effects. Images reproduced with permission [23].

Prior model: The main content of a prior model, also referred to as *prior knowledge*, can contain information of the heart shapes and important intensity features of the images. The common representations of prior models include the *atlas* [17, 19, 29, 67]

and the *deformable model* [11–15, 68]. In the atlas-based segmentation, one can construct an atlas pool, which contains atlases with different heart shapes, and use a multiple classification strategy to improve the segmentation [17, 24, 69, 70]. In the deformable model-based segmentation, the statistical shape model (SSM), also known as active shape model (ASM) [71], is commonly used for the segmentation of complex organs [72]. When an SSM also includes the statistical information of image intensity distributions, the model is also known as an active appearance model (AAM) [73]. The flexibility of an SSM plays an important role in accurate segmentation. As the variability of a statistical model is limited to the available training data, several techniques have been proposed to extend the flexibility of the SSM in the segmentation framework.

Fitting process: The fitting process requires a global localization to perform a deformable fitting. To improve the robustness, several studies proposed to use the hierarchical scheme, where the deformable fitting can be further decomposed into two stages: the initialization of substructures and the refinement of boundary delineation in detail [11–15, 19]. In the deformable fitting, shape regularization is required to maintain a realistic and smooth heart shape in the resultant segmentation.

3. METHODOLOGY

This section introduces the key techniques and analyzes the existing methodologies based on the classification of prior models and algorithms of fitting process. The works and the methodologies are summarized in Table 1 and Table 2, and the performance is presented in Table 3 and Table 4.

3.1. Prior Model

The prior model in the whole heart segmentation can contain information of the heart shape, the intensity distributions of the medical images, and sometimes the salient features of the landmarks and boundaries [9, 11–17, 19, 22, 62, 63, 68, 74, 75]. The representation of such prior knowledge includes the atlas and the deformable model using statistical shape model (SSM).

3.1.1. Atlas

An atlas is defined as an image pair consisting of an *atlas intensity image* and an *atlas label image*, as well as other useful prior information. Figure 1 shows an example of the atlas intensity image (Figure 1a) and atlas label image (Figure 1b). The atlas-based segmentation, referred to as *atlas propagation and segmentation*, employs image registration algorithms to fit the atlas image to the unseen image [19, 29, 46, 67, 69], as Figure 4 illustrates.

The atlas intensity image is generally constructed from an image or a set of images from the same modality as the expected unseen images, since it is used as the source image in the registration between the atlas and the unseen images. Original cardiac images can also be used as the atlas intensity images such as the cardiac CTA volumes [16, 17, 24] and the MRI volumes from the same patient [29]. This is also widely used in the brain image segmentation [76–79], where the brain MRI images are of high quality. In [19, 20], the atlas intensity image was constructed from the intensity mean image of a set of cardiac MRI images.

Key technique	Method	Reference
Existing Knowledge	Atlas	[16-20]
	Deformable model with statistical shape model	[9–15, 22]
Localization	Affine registration	[16–19, 24]
	Learning-based detection	[11-15, 24]
Initialization	Local affine registration of substructures	[18, 19, 24]
	Piecewise affine adaptation of substructures	[8, 11–13]
	Fitting control points of substructures	[14, 15]
Refinement	Voxel-based nonrigid registration	[16–19, 24]
	Boundary searching based on edge detection	[8–13]
	Boundary searching based on learning based	
	dense landmarks detection	[14, 15]
Shape Regularization	Smoothness penalty terms	[16–18]
	Combining smoothness penalty terms and the	
	shape-guided registration	[19, 20, 24]
	Constraining the deformation of prior model	
	within variability of statistical shape model	[9, 14, 15, 22]
	Shape-constrained deformable model	[8, 10–13]

Table 1. Categorization of fully automatic whole heart segmentation research

The atlas label image shown in Figure 1b is a label image with segmentation information of all the substructures of interests. The segmentation information will be propagated to the unseen image using the resultant transformation of the registration, to achieve the automated segmentation, as Figure 4 demonstrates.

In atlas propagation and segmentation, the segmentation propagation, i.e., the process of fitting the atlas to the unseen images, is achieved through a voxel-based registration method. In the registration, the atlas and the unseen image are regarded as the source space and the target space, respectively. The registration can use mutual information [80, 81] or normalized mutual information [82] as the similarity metric for cardiac images. Therefore, an atlas constructed from one modality or of one intensity distribution can potentially be used in the segmentation of images from the other modalities or of other intensity distributions.

To improve the robustness and accuracy of the registration, the atlas may also include other useful prior information, also known as *priors*. These priors can be the labels of other substructures of the heart, such as the descending aorta, or the adjacent organs (such as the liver), which may not be the interest of the segmentation itself. This information can be used to assist the registration, for example, to better initialize all substructures in the locally affine registration method [18, 19]. Other useful priors include the boundaries and anatomical landmarks. For example, the boundary information can be used to adaptively set the status of the control points in the free-form deformation registration, in order to improve the segmentation of the epicardium [19].

	0	•
		Method
Reference	Goal + Modality	Prior model + Localization + Initialization + Refinement + Shape Regularization
Lotjonen <i>et al.</i> 2004	Four chambers + multi- slice ^a cine MRI	SSM and mean image of AAM + Similarity registration + N/A + Radial FFD based registration + Deformation within shape variability of SSM
Koikkalainen <i>et al</i> .	Four chambers + multi-	SSM and mean image of AAM + Similarity registration + N/A + Search boundary
2008 [9]	slice cine MRI	and optimize eigen modes to fit SSM + artificially enlarged training set to increase variability of SSM
Peters et al. 2007	Four chambers and	Deformable model + Generalized Hough transform and similarity parametric
[12]	beginning trunks of	adaptation + piecewise affine parametric adaptation + search for optimal edge response
	aorta and pulmonary	along normal direction to mesh surface for parametric adaptation of vertices using
	artery + 3D MRI	simulated search + shape constrained deformable model to increase variability of the
		shape model
Ecabert et al. 2008	Four chambers and	Similar to Peters et al. 2007 [12]
[11]; 2011 [8]	beginning trunks of aorta	
	and pulmonary artery +	
	CT; Whole heart structure	
	and great vessels + CT;	
Peters et al. 2009	Four chambers and	Similar to Peters et al. 2007 [12]
[13]	beginning trunks of	
	aorta and pulmonary	
	artery + CT/3D MRI	
Zheng et al. 2008	Four chambers + CT	SSM and learning results of landmarks + Similarity transformation using marginal
[14, 15]		space learning and steerable feature + fit control points of substructure to corresponding
		landmarks detected using learning-based method + fit vertices of model to corresponding
		landmarks detected using learning-based method + project deformed model onto a
		subspace of SSM
		(Continued)

Table 2. Existing techniques for fully automatic whole heart segmentation

Existing techniques for fully automatic whole heart segmentation (<i>Continued</i>)	Method
Table 2. Existing te	

		Method
Reference	Goal + Modality	Prior model + Localization + Initialization + Refinement + Shape Regularization
Zhuang <i>et al</i> . 2010 [18-20]	Four chambers ^b + 3D MRI	Atlas (single or multiple) + Affine registration + Hierarchical locally affine registration + ACPS FFD registration + Smooth deformations and shape-guided registration
Zhuang et al. 2013	Four chambers and	Atlas (multiple) + Whole heart detection plus affine registration + Hierarchical locally
[24]	beginning trunks of aorta	affine registration + ACPS FFD registration + Smooth deformations and shape-guided
	and pulmonary artery + CT	registration
Kirisli et al. 2010	Four chambers and	Atlas (multiple) + Affine registration + N/A + FFD registration + Smooth deformations
[16, 17]	beginning trunks of	by including bending energy into the cost function of registration
	aorta + CT	
^a Multi-slice cine MRI: 7	The data included 4-5 slices from sh	ort-axis view and 4-7 slices from long-axis view, and were constructed as volumetric data using shape

based interpolation. ^bFour chambers: The segmentation results consisted of aorta and pulmonary artery as they were constructed in the atlas(es), but with the segmentation results of these great vessels not validated due to lack of gold standard.

		distance	(in mm) or/a	nd Dice score	(italicized)
Reference	Substructure	Whole Heart Accuracy	Run Time	Size of dataset Training Test	More information about data source and validation
Lotjonen et al. 2004 [22]	LV,EPI,RV,LA,RA	2.53 ± 0.70 mm	N/A	25 25	Multi-slice cine MRI, resolution of 1.0×1.0 or 1.4×1.4 mm in-plane and 6 to 7 mm slice thickness, from 25 healthy subjects used as training dataset as well test dataset.
Koikkalainen <i>et al.</i> 2008 [9]	LV,EPI,RV,LA,RA	$2.06 \pm 0.55 \text{ mm}$ 0.85 ± 0.05	700 seconds	24 25	Multi-slice cine MRI, resolution of 1.0×1.0 or 1.4×1.4 mm in-plane and 6 to 7 mm slice thickness, from 25 healthy subjects; leave-one-out cross validation.
Peters <i>et al.</i> 2007, 2009 [12, 13]	LV,EPI,RV,LA,RA, AO,PA	0.76 ± 1.08 mm	10-30 seconds	32 42	SSFP MRI, resolution of $0.5 \times (0.5 \text{ to } 0.7) \times (0.7 \text{ to} 0.9)$ mm, from 42 patients; 4-fold cross validation.
Ecabert <i>et al</i> . 2008 [11]	LV,EPI,RV,LA,RA, AO,PA	$0.82 \pm 1.00 \text{ mm}$	10-30 seconds	27 28	CT, resolution of $0.48 \times 0.48 \times (0.67 \text{ to } 3.0) \text{ mm}$, from 28 patient; leave-one-patient-out cross validation.
Ecabert <i>et al.</i> 2011 [8]	Whole heart structure and great vessels	0.70 ± 1.03 mm	12±1 seconds	35 37	CT, resolution of 0.28×0.28 to 0.49×0.49 mm in-plane and 0.33 to 0.50 mm slice spacing, 37 volumes from 17 patients as test dataset and another 35 subjects for training.
Zheng <i>et al.</i> 2008 [14, 15]	LV,EPI,RV,LA,RA	(1.13~1.57) ± (0.38~0.55) mm	4 seconds	242 323	CT, isotropic slices, resolution 0.28 to 0.74 mm in-plane and 0.4 to 2.0 mm slice thickness, from 137 patients; four-fold cross validation.

Table 3. Validation results of the existing whole heart segmentation methods with accuracy reported in surface-to-surface

		uistance (III III		re scole (Itali	
Reference	Substructure	Whole Heart Accuracy	Run Time	Size of datase Training Tes	t More information about data source and validation t
Zhuang <i>et al.</i> 2010 [19]	LV,MYO,RV,LA,RA	1.47 ± 1.55 mm 0.84 ± 0.05	2-4 hours	10 37	SSFP MRI, acquired resolution of $2 \times 2 \times 2$ mm, 10 healthy subjects for atlas construction and 37 subjects (18 healthy and 19 patients consisting of 9 pathologies) for test dataset.
Zhuang <i>et al.</i> 2010 [20]	LV,MYO,RV,LA,RA	0.912 ±0.16	N/A	20 21	SSFP MRI, acquired resolution of $2 \times 2 \times 2$ mm, from 21 healthy subjects; leave-one-out cross validation.
Zhuang <i>et al</i> . 2013 [24]	LV,EPI,RV,LA,RA, AO,PA	$1.17 \pm 1.39 \text{ mm}$ 0.89 ± 0.02	231 seconds	14 15	CT, from 15 patients; leave-one-out cross validation.
Kirisli <i>et al.</i> 2010 [16, 17]	LV,EPI,RV,LA,RA, AO	$0.99 \pm 1.25 \text{ mm}$ $0.89 \sim 0.95$	20 minutes	7 8	CT, resolution of $0.32 \times 0.32 \times 0.4$ mm, from 8 patients; leave-one-out cross validation.

ccuracy reported in	
e heart segmentation methods with a	or/and Dice score (italicized)
ilts of each substructure in existing whole	surface-to-surface distance (in mm) o
Table 4. Validation resu	

Reference	LV	EPI/MYO	RV	LA	RA	AO	PA
Lotjonen <i>et al</i>				0000-720		A 114	A L L
2004 [22]	$2.01 \pm 0.31 \text{ mm}$	$2.77 \pm 0.49 \text{ mm}$	2.37 ± 0.50 mm	$2.50 \pm 0.88 \text{ mm}$	$2.95 \pm 1.50 \text{ mm}$	N/A	N/A
Koikkalainen	$1.46 \pm 0.32 \text{ mm}$	$1.93 \pm 0.68 \text{ mm}$	$2.19 \pm 0.32 \text{ mm}$	$2.45 \pm 0.82 \text{ mm}$	$3.15 \pm 1.48 \text{ mm}$	N/A	N/A
et al. 2008 [9]	0.91 ± 0.03	0.91 ± 0.03^{a}	0.85 ± 0.03	0.74 ± 0.09	0.68 ± 0.14		
Peters et al.							
2007, 2009							
[12] [13]	$0.69 \pm 1.13 \text{ mm}$	$0.83\pm1.17~\mathrm{mm}$	$0.74 \pm 0.96 \text{ mm}$	$0.72 \pm 1.14 \text{ mm}$	$0.63\pm0.95~\mathrm{mm}$	$0.60\pm0.70~\mathrm{mm}$	$0.73 \pm 1.11 \text{ mm}$
Ecabert et al.							
2008 [11]	$0.77 \pm 1.14 \text{ mm}$	$0.68\pm0.96~\mathrm{mm}$	$0.63\pm0.66~\mathrm{mm}$	$0.70\pm0.87~\mathrm{mm}$	$0.82\pm1.00~\mathrm{mm}$	$0.74\pm0.89~\mathrm{mm}$	$0.83\pm1.05~\mathrm{mm}$
Ecabert et al.							
2011 [8]	$0.77 \pm 1.14 \text{ mm}$	$0.68\pm0.96~\mathrm{mm}$	$0.63 \pm 0.66 \text{ mm}$	$0.70 \pm 0.87 \text{ mm}$	$0.82\pm1.00~\mathrm{mm}$	$0.60 \pm 1.14 \text{ mm}$	$0.50\pm0.49~\mathrm{mm}$
Zheng et al.							
2008 [14, 15]	$1.13 \pm 0.55 \text{ mm}$	$1.21 \pm 0.42 \text{ mm}$	$1.55\pm0.38~\mathrm{mm}$	$1.32 \pm 0.42 \text{ mm}$	$1.57\pm0.48~\mathrm{mm}$	N/A	N/A
Zhuang et al.	$1.06 \pm 1.02 \text{ mm}$	$1.69 \pm 1.60 \text{ mm}$	$1.50\pm1.51~\mathrm{mm}$	$1.69 \pm 1.68 \text{ mm}$	$1.51 \pm 1.62 \text{ mm}$	N/A	N/A
2010 [19]	0.92 ± 0.02	0.77 ± 0.06^{b}	0.87 ± 0.04	0.81 ± 0.10	0.84 ± 0.05		
Zhuang et al.							
2010 [20]	0.949 ± 0.014	0.887 ± 0.025^{b}	0.930 ± 0.021	0.883 ± 0.026	0.891 ± 0.038	N/A	N/A
Zhuang et al.							
2013 [24]	$0.90 \pm 0.78 \text{ mm}$	$1.09 \pm 1.01 \text{ mm}$	$1.12 \pm 1.16 \text{ mm}$	$1.67 \pm 1.96 \text{ mm}$	$1.70\pm1.77~\mathrm{mm}$	$0.73\pm0.81~\mathrm{mm}$	$1.14\pm1.55~\mathrm{mm}$
	0.94 ± 0.03	0.83 ± 0.05^{b}	0.91 ± 0.03	0.87 ± 0.04	0.85 ± 0.05	0.91 ± 0.03	0.80 ± 0.11
Kirisli et al.							
2010 [16, 17]	$0.62\pm0.63~\mathrm{mm}$	$1.04 \pm 1.15 \text{ mm}$	$1.40\pm1.47~\mathrm{mm}$	$0.66 \pm 0.84 \text{ mm}$	$1.44 \pm 1.88 \text{ mm}$	$0.44 \pm 0.60 \text{ mm}$	N/A
	0.95	0.95^a	06.0	0.94	0.89	0.94	

^{α}The volume compared in the Dice includes the blood pool and the myocardium of the left ^bThe volume compared in the Dice is the myocardium of the left ventricle.



Figure 4. Workflow of the atlas propagation and segmentation.

3.1.1.1. Atlas Selection and Multiple Atlas Strategy

In atlas propagation and segmentation, it is important to construct atlases with high quality and to select the best atlas(es) for segmentation propagation. An ideal atlas not only should have a high quality intensity image, but also should have a heart shape similar to the unseen image. This is practically important and represents one of the major challenges for automatic processing of cardiac images, as discussed in Section 2. Since an atlas does not naturally contain the statistical information of shape variations, a solution proposed was the multiple atlas strategy [17, 69, 70]. The multiple atlas strategy selects a set of atlases and uses each of the selected atlases to segment the unseen image with a segmentation result. The final segmentation is a fusion from these results using an algorithm such as the majority voting approach [83]. The multiple atlas strategy is in particular efficient and effective when the medical images with manual labeling can be directly used as atlases. For example, in the cardiac CT/CTA segmentation and brain MRI segmentation, the original medical images are of high image quality and can be employed as the atlas intensity images [16, 17, 24, 29]. This method is referred to as the multiple atlas propagation and segmentation (MAPS). In whole heart segmentation of cardiac MRI volumes, MAPS did not demonstrate convincing improvement compared to the single atlas propagation and segmentation (SAPS) using a mean intensity atlas [19]. It was proposed to use multiple path propagation and segmentation (MUPPS) to achieve the multiple classifications [20]. In MUPPS, the common atlas was constructed from the mean of a set of training images, including the mean of intensity and the mean of shapes. By registering the atlas label image to the corresponding label images of the training data, a set of transformations were determined. These transformations were used as priors and referred to as *paths* [20]. Each transformation could be employed to deform the common atlas, including the intensity image and the label image, to generate a new atlas if this transformation was selected in the path-ranking process. The new atlases had the same intensity distributions as the common atlas, but with different heart shapes defined by the prior transformations. Therefore, MUPPS resembles the MAPS except that the atlases were generated using the prior transformations, instead of the original cardiac MRI images. In both MAPS and MUPPS, an atlas/path ranking scheme should be adopted to efficiently select a small number of "good" atlases from the atlas pool which generally has a large number of candidates. The inspiration from MUPPS is



Figure 5. Example of generated atlas using MUPPS techniques. (a) The common atlas. (b) The pathological image showing a hypertrophic right ventricle. (c) The generated atlas from the common atlas with right ventricle hypertrophy.

that one can generate new atlases from existing ones, instead of requesting more training samples to construct new atlases. Figure 5 shows an example of a new atlas with right ventricle hypertrophy generated using such techniques [20].

3.1.2. Statistical Shape Model for Deformable Model-Based Segmentation

The deformable model was first introduced to the computer vision field and had gained its popularity since late 1980's, in particular after the "snake" method was proposed [84, 85]. Deformable model-based segmentation employs shape priors to maintain an optimal shape in the resultant segmentation. The statistical shape model (SSM) [71] is one of the most investaged methods to build the statistical information of the shape priors. In the SSM-based segmentation, the deformation of the model is generally constrained to be within or close to the shape manifold defined by the SSM. These segmentation approaches combine the advantages from both the SSM and the deformable model-based segmentation framework, and they are more suitable for the segmentation of organs with complex shapes [72].

The SSM, originally known as the active shape model (ASM) [71], has been widely investigated and used in modeling and segmentation of complex objects in the medical image computing field [72]. Compared to the atlas, the SSM has the advantage of being able to build in the statistical information of shapes of the heart. The procedure of constructing an SSM for whole heart segmentation includes the following steps, as outlined in Figure 6:

• Represent images using a shape model. Cootes *et al.* employed the point distribution model (PDM) [71]. The connection between the dense points can be formed to create a surface mesh model for a deformable model-based segmentation [9, 11–15, 25, 34]. The shape model embeds the segmentation information and will be propagated to the unseen images. Therefore, it is important to choose a representation, such as the extracted volume or surface of a



Figure 6. Flowchart of constructing statistical shape model.

substructure, from which the segmentation information can be efficiently interpreted. To this end, the triangulated surface mesh model has been preferred and commonly adopted for the whole heart segmentation in the SSM-based methods [11–15].

- Align image coordinates to normalize the shapes of the training data. To build an SSM, the training data need to be aligned to a common space, as they can be in different coordinate systems and with different orientations and sizes. By aligning the training data to this common space, the variations that are not of interests to the resultant SSM are then excluded from modeling.
- Find correspondence between each training shape. To construct the covariance matrix, one needs to find the correspondence of each vertex between two training shapes or between a training shape and the common space. Peters *et al.* [12, 13] and Ecabert *et al.* [11] employed a common mesh model and deformed this mesh model to each training image to extract its shape. The correspondence between the vertices of the common model and those of the training shapes was then automatically formed. Zheng *et al.* [14, 15] allocated vertices to the surfaces of the manually segmented whole heart volume of each training case using a well defined protocol. Each of the vertices was indexed and then used to form a triangulated mesh model. Since the number and relative positions of the vertices to the whole heart surface were the same for all training data, the correspondence of each training case was then defined implicitly by the indices of the vertices.
- Construct mean shape and covariance matrix. Given the shape of a training case is represented by a set of *n* vertices as $X_i = \{v_1, v_2, \dots, v_n\}$, the mean shape of *N* training data is computed as follows:

$$\overline{X} = \frac{1}{N} \sum_{i=1}^{N} X_i, \tag{1}$$

and the covariance matrix is given by

$$C_{\text{data}} = \frac{1}{N} \mathbf{A} \times \mathbf{A}^{T}, \text{ where } \mathbf{A} = \left[\mathbf{X}_{1} - \overline{\mathbf{X}}, \mathbf{X}_{2} - \overline{\mathbf{X}}, ..., \mathbf{X}_{N} - \overline{\mathbf{X}} \right].$$
(2)

 Apply PCA to compute SSM. Applying the eigen decomposition to the covariance matrix C or singular vector decomposition (SVD) to A results in ordered eigen values, λ_1 , λ_2 , ..., λ_{dN} ($\lambda_k \ge \lambda_{k+1}$) where *d* is the dimension of the vertex coordinates, and the corresponding eigen vectors, V_1 , V_2 , ..., V_{dN} , are known as variation modes or eigen modes of the SSM. A valid shape can be estimated from these modes, as follows:

$$\boldsymbol{X}_{\text{new}} = \overline{\boldsymbol{X}} + \sum_{k=1}^{N'} b_k \boldsymbol{V}_k, -3\sqrt{\lambda_k} \le b_k \le 3\sqrt{\lambda_k}, \qquad (3)$$

where $\{b_k\}$ are the weights for the first N' eigen modes which are the most significant variation modes in the shape space.

3.1.2.1. Feature Priors

To employ a deformable model for automatic segmentation, the feature priors should also be constructed in the prior model. These feature priors are used in the model fitting process to align the model elements to the ground truth positions in the unseen images. Three methods have been reported to construct and apply these feature priors.

The first method is to learn the texture and location features of each vertex using a learning-based algorithm, such as the steerable features and Adaboost-based classifiers [14, 15]. The feature priors of each vertex are the resultant feature bank which forms a strong classifier to detect the corresponding position in unseen images. These approaches are computationally efficient in detecting landmarks which are defined to the vertices of the model, though the training stage can be computationally expensive and memory demanding.

The second method is the simulated search approach [11–13]. In this approach, a locally optimal boundary detection function was assigned to each mesh triangle in the deformable model. The optimal functions were determined in the model training phase, where the performance of any boundary detection function was evaluated for each element of the model. As a result, the boundary detection can be optimized for each element during the model training. This simulated search was demonstrated to be able to identify the suitable classes of features when a new segmentation task is presented and to be applicable to multi-modal cardiac image segmentation using a single algorithmic framework [13].

The third method constructs texture variations and combines it with the SSM to form a statistical appearance model, which is referred to as active appearance model (AAM) [73]. The construction of AAM is similar to that of SSM. An AAM which has statistical information of both the shapes and the texture patterns enables employing the voxel-based registration to fit the AAM to an unseen image. The parameters to be optimized are the weights assigned to the eigen modes [40]. However, the dimension of vectors representing the texture patterns of a whole heart image can be very large, and applying PCA to the covariance matrix can be practically challenging. Therefore, Lotjonen *et al.* and Koikkalainen *et al.* used the mean intensity image of the AAM for voxel-based registration [9, 22]. The mean intensity image of the AAM is equivalent to the atlas intensity image constructed from the mean of the training data in [19, 20].

3.1.2.2. Extending Flexibility of Statistical Shape Model

In deformable model-based segmentation, the flexibility of an SSM plays the key role in high-performance segmentation. As the SSM only represents the shape variations of the training data, the variability of the SSM is restricted and the accuracy of segmentation is limited if the shape of an unseen image is very different from the training samples [22]. Since it is practically demanding to obtain a large number of training data, several techniques have been proposed to extend the flexibility by either artificially enlarging the training set or releasing the shape constraint of the SSM.

Koikkalainen *et al.* [9] investigated five approaches to artificially enlarge the training set for an SSM, including the *Nonrigid Scaling* method [86], the *Nonrigid Movement* method [86], the *Smoothing* method [87], the *PCA & FEM* method [88], and the *Partitioning* method [89, 90]. The *Nonrigid Moving* method and the *PCA & FEM* method may be the two most promising techniques in the whole heart segmentation framework. The *Biomechanical Model* method [91] and the *Shape Constrained Deformable Model* method [68], which were not covered in [9], also demonstrate potentials of effectively extending the flexibility of an SSM.

Let $\{X_i\}$, i = 1...N be the set of N training cases after shape normalization using, e.g., the Procrustes alignment [92], and \overline{X} be the mean shape of the training set; let $v_j(i), j = 1...n$, be the *j*th vertex of the training case X_i and \overline{v}_j be the mean coordinate of all the *j*th vertices in the training set.

The *Nonrigid Scaling* method [86], extending the original idea of the adaptive focus method in [93], scales the surface inside a deformation sphere to generate new samples. The new coordinate of a vertex is given by

$$\hat{\upsilon}_{j} = \upsilon_{j} + t_{j} \cdot (\upsilon_{j} - \overline{\upsilon}_{j})$$

$$t_{j} = \begin{cases} \frac{e^{-2||\upsilon_{j} - c||^{2}/r^{2}} - e^{-2}}{1 - e^{-2}}, & \text{if } ||\upsilon_{j} - c|| < r \\ 0, & \text{else} \end{cases}$$
(4)

where t_j is the scaling factor, and *c* and *r* are the center and radius of the sphere, respectively. The center of the deformation sphere is randomly positioned on surfaces of the training samples. By deforming the surfaces of the original training data using eqn. (4), new training samples are generated.

The Nonrigid Movement method [86] is similar to the Nonrigid Scaling method:

$$\hat{v}_{j} = v_{j} + d_{j}$$

$$d_{j} = \begin{cases} \frac{e^{-2\|v_{j} - c\|^{2}/r^{2}} - e^{-2}}{1 - e^{-2}}u, & \text{if } \|v_{j} - c\| < r \\ 0, & \text{else} \end{cases}$$
(5)

where *u* is a random vector and its elements are normally distributed within [-r/2 ... r/2]. Koikkalainen *et al.* used 31 spheres and set r = 50 mm for both the *Nonrigid*

Scaling method and the *Nonrigid Movement* method, generating (31*N*+*N*) samples for the training of SSM.

The *Smoothing* method [86, 87] combines the statistical variation modes of the original training data and the smoothness modes by constructing a mix covariance matrix:

$$C_{\rm mix} = C_{\rm data} + \beta \cdot C_{\rm smooth}, \tag{6}$$

where C_{smooth} is the smoothness covariance matrix and $\beta = \frac{(dn)^2}{N-1}$. The elements of C_{smooth} are set to 1 if they are corresponding to identity vertices, set to 0.5 if they are corresponding to neighboring vertices [87] or vertices belonging to the same triangle [86], and set to 0 for else. C_{smooth} represents synthetic variations to the subsequent SSM.

The *PCA & FEM* method [88] calculates the vibrational modes of elastic materials. These vibrational modes are used to generate synthetic variations and added to the original training shapes to generated new samples:

$$\hat{\boldsymbol{X}}_i = \boldsymbol{X}_i + \boldsymbol{\Phi}_i \cdot \boldsymbol{b},\tag{7}$$

where Φ_i is the matrix of eigenvectors representing the modes of vibration about training example X_i , and **b** is a weight vector. Similar to the *Smoothing* method, the *PCA & FEM* method can also include the synthetic variations in the resultant SSM by constructing a mix covariance matrix:

$$C_{\text{mix}} = C_{\text{data}} + \gamma \cdot \frac{1}{N} \sum_{i=1}^{N} \Phi_i \Lambda_i \Phi_i^T, \qquad (8)$$

where Λ_i is a diagonal matrix consisting of the corresponding eigen values of the stiffness matrix [88].

The *Partitioning* method, including the hierarchical ASM [89] and the partitioned ASM [90], divides the surface model of a training image into N_P patches, $\{P_k\}$, k = 1, ..., N_P . The rationale is to separate the local shape variations of a patch from the global shape and specifically model such local variations within the subsequent SSM. This can be done by constructing an individual SSM for each patch [80, 81], or achieved by generating more training data, which only exhibit the shape difference in the patch, for the SSM [9]. The latter method was tested in the whole heart segmentation. For each training case X, N_P new samples can be simulated by keeping the vertices of one patch and replacing the other patches using the mean model:

$$\hat{X}_{i}(k) = \begin{cases} v_{j}(i), \quad v_{j}(i) \in P_{k} \\ \overline{v}_{j}, \quad v_{j}(i) \notin P_{k} \end{cases}.$$
(9)

The enlarged training set, including the original training data, then has N(P+1) samples.

The *Biomechanical Model* method [91], to the best of our knowledge, has not been applied to the SSM study of cardiac modeling. The simulated training shapes using the methods mentioned above do not necessarily represent a realistic shape of the heart, although the methods can extend the flexibility of the resultant SSM. For example, the PCA & FEM method only add vibrational modes of elastic materials to the resultant SSM, while the elasticity does not fully represent the complex tissue property of an organ such as the heart. Davatzikos et al. developed a framework for modeling anatomical deformations using a biomechanical model [91]. The emphasis of the work was on the function of predicting anatomical changes due to surgical planning or tumor growth. This idea was subsequently extended to estimate intra-operative deformations of the prostate [94]. Hu et al. employed a biomechanical model to simulate the deformations of soft-tissues caused by the random placements of an ultrasound probe [95]. The deformations were then applied to a patient-specific surface model constructed from an MRI image to generate new samples for the training of an SSM of the prostate. The patient-specific SSM was used to register the interventional 3D ultrasound images for the fast nonrigid alignment of the pre-operative MRI and the interventional ultrasound.

The *Shape Constrained Deformable Model* method [11–13, 68, 93, 8] does not artificially enlarge the training set; instead, it relaxes the constraint from the SSM by incorporating the deformable shape model in a regularization term of the energy function in the active contour segmentation framework:

$$\arg\min \quad \boldsymbol{E} = \boldsymbol{E}_{\text{ext}} + \alpha \boldsymbol{E}_{\text{int}}.$$
 (10)

In this framework, E_{ext} is the external energy measuring the distance between the deformable model to the detected boundaries in the unseen image, and E_{int} is the regularization term which penalizes the deformations when the surface model is distant from the hyper space of the SSM or simply the mean shape model [11–13]:

$$\boldsymbol{E}_{\text{int}}(\boldsymbol{V}, \boldsymbol{P}, \boldsymbol{T}) = \sum_{j=1}^{n} \sum_{a \in N(j)} \left[(v_j - v_a) - \boldsymbol{T} \left(\left[m_j \right] - \left[m_a \right] \right) \right]^2, \tag{11}$$

where $V = \{v_i\}$ are vertices of the deforming mesh model, $\{m_i\}$ are the corresponding vertices in the deformable mesh model, P are parameters weighting the eigen modes of the PCA model, N(i) are the indices of the neighbor vertices of vertex v_i in the mesh model, and T is a geometric transformation. The parameters P can disappear if the mean shape model is used. Given the corresponding vertices in the epicardial and endocardial surfaces also be connected in the mesh model, eqn. (11) also implements the penalty term to maintain the connection distance between the endocardium and epicardium [25]. The piecewise affine transformation model is used to model the geometric transformation T [11–13]. The piecewise affine transformation model, which can be implemented in a hierarchy fashion, further extends the flexibility of the shape constrained deformable model and improves the capability of modeling local details.

3.1.3. Other Potential Modeling Techniques

Modeling the anatomy of the heart can be challenging as the heart has a complex structure. For segmentation propagation, a number of simplified representations of the shape model have been used, including manually delineated contours, mathematical models, point distribution models, surface mesh models, volumetric models with elastic material or bio-mechanical properties, and the sparse shape composition.

Hautvast *et al.* semi-automatically segmented the ventricles and myocardium of cine cardiac MRI by propagating the manual delineation of a selected phase to other phases [74]. The prior model was the initial manual delineation.

Mathematical models, such as a conic equation, were used to model the ventricle cavity as an ellipsoid [32]. This model is simple in construction and can be elegantly formulated in mathematics for deriving the internal energy. However, modeling the whole heart structure can be too complex for this method. Medial representation (M-Reps) [96] is a more sophisticated tool to model objects with irregular shapes. The applicability was demonstrated in image segmentation for several applications in [97], except for cardiac images. Pilgram *et al.* built a heart model based on M-Reps which may be applied to this task [98]. However, its application to whole heart segmentation, to the best of our knowledge, has not been reported.

The point distribution model [88] provides dense point cloud on the surfaces of the modeled objects. For segmentation propagation, the connection between these points is generally formed to create a surface mesh model, for example, for the segmentation of the ventricles [25, 34] and the whole heart [9, 11–15]. The volumetric models with built-in material property, such as nonlinear elasticity [75] or biomechanical property [99], may also be applicable to whole heart segmentation. Constructing these models is one challenging task, while efficiently solving the deformation fields from the applied forces is the other.

Zhang *et al.* [100, 101] proposed a sparse shape representation to approximate the shape of an unseen image using sparse linear combination of other shape instances. These shape instances had been previously annotated. Without using statistical shape decomposition methods, they employed the sparsity constraint to derive a smooth and realistic shape. This sparse shape representation method has been used in the liver segmentation where the objective organ can be considered as one simple structure. In whole heart segmentation, the objective organ consists of several substructures which need to be separated from each other. Hence, it can be a challenge to apply the method directly. More details can be found in the review papers for cardiac modeling [3] and shape representation and description techniques [102].

Prior models are generally embedded in the segmentation algorithms and determine the detail and representation of the segmentation results. Therefore, the content of anatomical modeling of the heart can be different for different applications. For example, both the atrioventricular valves and the virtual valve planes [14, 15] can be modeled to define the separating boundaries between the atria and ventricles. The segmentation of ventricular cavities can either include or exclude the papillary muscle, depending on the definition of the prior model. Most of the reported cardiac segmentation articles modeled the smooth endocardial surfaces without considering the details inside the heart, although these finer structures, such as the trabeculae, can be captured and delineated using the state-of-the-art CT scanners [103].

3.2. The Fitting/ Registration Process

A prior model enables fitting the model to an unseen image for segmentation propagation. In general, the shape difference between the model and the unseen image is nonrigid, so that the global rigid or affine transformations are not sufficient for the registration between them. Therefore, the fitting process should include a global affine registration to correct the global difference such as the position, size, and orientation, and then a local deformable registration to correct the local difference. To achieve robust and accurate segmentation propagation, the deformable registration is generally implemented in a coarse-to-fine fashion. Therefore, the fitting process is considered as a hierarchy scheme, consisting of the three stages: localization of the whole heart, initialization of substructures, and refinement of boundary delineation [11–15, 19]. In the deformable registration, shape regularization is needed. The flowchart of the fitting process is shown in Figure 7. The sections below introduce the methodologies of the following four key techniques that are used to classify published works (Table 1 and Table 2):

- *Localization* of the whole heart, including location, orientation and size, etc.
- *Initialization* of substructures of the heart.
- *Refinement* of the boundary delineation in detail.
- *Regularization* of Shapes to maintain a realistic and smooth heart shape in the resultant segmentation.

3.2.1. Localization

The goal of the localization is to align the global shape of the prior model to that of the unseen image. Therefore, this localization stage can correct global mismatch of translation, orientation (rotation), size, and sometimes shearing as well between the model and the unseen image. An affine transformation can be used to model this difference. Since the shearing difference is not evident and thus is less important, a *similarity* transformation is commonly adopted to correct the translation, rotation, and size difference. There are two major techniques reported for the localization task in the automatic whole heart segmentation, including intensity-based image registration and learning-based feature detection.



Figure 7. Flowchart of fitting the prior model to unseen data for segmentation propagation.

The global affine registration has been well developed in the past two decades [104]. It is commonly used for the localization in the atlas propagation and segmentation frameworks [16–19]. The mean image of the AAM was employed to register to unseen images for localization [22, 9].

Learning-based detection methods train and record the common and discriminative features in cardiac images using a large number of training data. Haar-like features are used to train a detector to find the center of the heart [14, 15]. To estimate the orientation and size, Zheng et al. proposed the steerable features which can be efficiently computed without rotating or scaling the unseen images [14, 15]. Also, they further proposed the marginal space learning (MSL) to gradually increase the dimension of the parameter space. MSL significantly reduces the number of testing hypothesis, of which each corresponds to one case in the quantized similarity transformation parameter space. Weese et al. employed the generalized Hough transform to localize the center of the heart and estimate its size using a set of quantized isotropic scaling parameters [11-13]. To further perform the localization with orientation and anisotropic scaling, they modified the surface mesh model with a global affine constraint. Both the localization methods in Zheng et al.'s and Weese et al.'s works registered their prior models to an unseen volume using a similarity transformation. In general, the learning-based localization approaches are computationally efficient and robust in searching the center and size of the heart. To estimate an affine transformation, Shi et al. [38] and Zhuang et al. [24] employed the Haar-like feature training-and-detection method to quickly estimate the center and size of the heart, and then applied an affine registration to further correct the difference for their cardiac ventricle segmentation.

3.2.2. Initialization

Initialization of substructures of the heart is an important stage to bridge the global heart localization and the follow-up refinement of local details. It is also the key technique to achieve the robust whole heart segmentation for a wide range of pathological cases with large variations of shapes.

The locally affine registration method (LARM) was proposed for the initialization of substructures in [18, 19]. To apply LARM, one needs to define a set of local regions or landmarks. Each of them is assigned a locally affine transformation. LARM has a mechanism to maximally maintain the affinity of the pre-defined local regions by simultaneously optimizing all the locally affine transformations. It produces diffeomorphic resultant transformation to preserve the topology of the prior model after the propagation. An inverse deformation field can be efficiently computed using the proposed dynamic resampling and distance weighting interpolation scheme, which theoretically can produce an inverse transformation with a maximal error less than the voxel size of the reference image. The initialization was implemented in three stages, a coarse-to-fine hierarchy framework [19]:

• In the first stage, only two local affine transformations are performed. The corresponding local regions are defined to the atrial region, including the two atria and the great vessels, and the ventricular region, including the two ventricles.

- In the second stage, the right ventricle and the right atrium are separated from the regions defined in the first stage to add another two local affine transformations.
- Finally, seven local regions, including two ventricles, two atria, pulmonary artery, ascending aorta, and descending aorta, are used to define a seven-local-region LARM.

In all three stages, the atlas image was used as the target image. Therefore, an inverse transformation from the unseen image to the atlas was also computed.

Weese *et al.* [11–13] proposed the piecewise affine model for the initialization process. In the piecewise affine adaptation, each piece of the pre-defined substructures was assigned an affine transformation. The change of the shape of the substructure was constrained within the local affine transformation. The adaptation was also implemented in a hierarchy scheme as follows:

- First, a global affine transformation was used for the adaptation of the whole heart.
- Each of the anatomical substructures, such as the four chambers and the great vessels, was then assigned an affine transformation.
- Finally, the anatomical substructures could be further divided into finer substructures, and more pieces of affine transformations were assigned to them.

The realization of substructure initialization was embedded into the model deforming and shape regularization process [14, 15]. After the localization, they employed the learning-based landmark detection to detect a small set of control points of each substructure. The control points are mainly scattered to the separating positions and the skeleton of the substructures, such as the valve annulus and the ventricular divergence planes. These control points not only can estimate the boundaries between the atria and ventricles, where there is no distinct boundary from the intensity distributions of the CT or MRI volumes, but also can initialize the global shape of the four chambers. To regularize the global shape of the model after the control point adaptation, the deformed model was projected onto a shape subspace with 25 dimensions (first 25 eigen modes) of the prior SSM.

Finally, it should be noted that the initialization stage can be optional in a whole heart segmentation method if other techniques are employed to guarantee the substructures to be also well initialized after the global localization stage [9, 16, 17, 22]. The multiple atlas strategy was used in [16, 17], the training data set was artificially enlarged to extend the shape variability of the SSM [9], and an SSM whose shape variations covered the test data (the test data were the training data) was constructed for the segmentation [22].

3.2.3. Refinement

Refinement is the final stage of the fitting process. It produces more local deformations to the model and can achieve better accuracy compared to the previous processes.

In atlas propagation and segmentation, nonrigid registration is used to refine the final propagation. Kirisli *et al.* [16, 17] employed the classic FFD registration [105] which uses the bending energy as the penalty term to guarantee the smoothness of the resultant

deformations. Similar to other generically applicable nonrigid registration methods, such as the fluid registration [106], the registration in these works does not use the prior knowledge in the atlas to guide the registration. This unsupervised refinement is however prone to generate erroneous results, particularly in the segmentation of myocardium [19]. Figure 2c shows an example that the epicardium was erroneously delineated as the adjacent liver tissue using the classic FFD registration. The FFD with active control point status (ACPS FFD) was proposed to incorporate the shape of the heart and guide the nonrigid registration [19, 20].

Deformable model-based methods refine the final segmentation by searching boundary/edge points in the unseen image, and then locally adapting the mesh model to align the vertices or barycenter of triangles of the model to the detected corresponding points [9, 11-15]. Koikkalainen et al. adopted the method of Cootes et al. [71, 73] in the fitting process. This method searches the optimal weights for the eigen modes of the SSM to minimize a cost function, which measures the difference between the texture built in the SSM and the corresponding texture in the unseen image. Peters et al. employed the trained boundary detectors to detect the boundary points and refined the final segmentation by minimizing the cost function in eqn. (11) [11–13]. The detection functions were selected (trained) in the model training stage using the simulated search approach, and the optimized parameters included the piecewise affine transformations, the weights of eigen modes of the SSM, and the vertex coordinates. These parameter sets were optimized by iterating two steps: first to optimize the transformation parameters, and then to optimize the weights of eigen modes and vertex coordinates. Zheng et al. [14, 15] searched boundary points, as well as other salient landmarks for initialization, using learning-based point detection algorithm. The refinement step searched all the boundary points which were predefined according to the anatomy of the heart and were trained using steerable features. The detection of a boundary point was guided such that the searching range of the point was constrained within a region initialized by the corresponding point in the deformed surface model.

3.2.4. Shape Regularization

The shape regularization is crucial in automatic segmentation of the organs with complex shapes and multiple components. In general, it is embedded into the deformable fitting processing, including the initialization and refinement stages, to maintain the global shape. The objective of the shape regularization is to maintain the deformed model with smooth surfaces, a realistic heart shape, and robust propagation of indistinct boundaries which are defined according to the anatomical knowledge.

In atlas propagation and segmentation, the shape regularization is mainly handled with the original shape of the atlas and the deformation regularization process in the registration. Kirisli *et al.* [16, 17] used the bending energy implemented in the classic FFD registration [105] as the shape constraint. Zhuang *et al.* [18, 19] used the LARM and the ACPS FFD for the registration processes in the initialization and the refinement stages, respectively. Both methods adopted a hierarchical scheme and incorporated the shape of the heart in the atlas to guide the registration process. This shape-guided registration was shown to improve the performance of the segmentation propagation [19].

In the SSM based segmentation, the widely used approach is to constrain the deformation of the shape model within the variability of the SSM [9, 14, 15, 28, 40]. The disadvantage is that the accuracy of the segmentation is limited by the variability of the SSM, namely the training dataset. Zheng et al. [14, 15] regularized the shape of the model in each nonrigid estimation step by projecting the deformed model onto a subspace of the SSM, which is equivalent to using the first few eigen modes to estimate the new shape. Koikkalainen et al. [9] adopted the fitting method of Cootes et al. [71] to limit the weights of the eigen modes within three times of the standard deviation, but extended the variability of the SSM by artificially enlarging the training set [9]. Peters et al. [11-13] employed the shape constrained deformable model to regularize the shape of the segmentation result. This method relaxed the constraint from the SSM by incorporating the deformable model in a regularization term of the energy function in the active contour segmentation framework. In the multi-stage adaptation process, a mean mesh of the whole heart structure could be used and matched to unseen images, where no variation mode of the SSM was needed in the fully deformable adaptation thanks to the piecewise affine transformation model.

4. VALIDATION ISSUES

The results of whole heart segmentation can either be presented as a binary volume or a surface mesh for each substructure, depending on the requirement of clinical applications. For example, for computing functional indices such as myocardium mass or ejection fraction, the binary volume is preferred, while for 3D rendering of the heart structure, the surface mesh is more useful. The volumetric and surface presentations are also convertible to each other using existing techniques. To objectively evaluate a whole heart segmentation algorithm, three issues need to be considered.

Firstly, a large amount of clinical data, with a wide range of pathologies, are needed. Since the major challenge of automated whole heart segmentation stems from the large shape variation of the heart, this test dataset should cover the different shapes that the clinical data would have in practice.

Secondly, a ground truth or gold standard segmentation of each test image is required. In general, manual delineation is used to construct gold standard segmentation. This manual work can combine semi-automatic tools with friendly and efficient user interface to interactively correct the delineation. Also, the definition of boundaries of substructures should be consistent for both the prior model and the gold standard of all test data. For example, the blood cavity of the left ventricle can either include or exclude the papillary muscle, and the boundary between the left ventricle and atrium can be defined using either the actual position of the mitral valve or a flat plane positioning on the annulus of the mitral valve. Similar situation applies to the right ventricle and right atrium. For the aorta and pulmonary artery, since they extend to other organs of the body and the field of view may not cover the whole structure, the validation of whole heart segmentation can focus on the major trunk of the arteries as discussed in the Introduction section (Section 1).

Finally, objective evaluation protocols should be employed, particularly for cross study comparisons. For a binary volume of a substructure, one can use the Dice score [107] or Jaccard index [108] to evaluate the similarity between the segmentation

volumes from automatic segmentation V^{auto} and the gold standard V^{gd} . The generalized Dice score and Jaccard index [109], which summarize the overlap of multiple labels, can be used to assess a whole heart segmentation result [20]:

$$g\text{Dice} = \frac{2\sum_{\nu=1}^{N_{\nu}} \left| V_{\nu}^{\text{auto}} \cap V_{\nu}^{\text{gd}} \right|}{\sum_{\nu=1}^{N_{\nu}} \left| V_{\nu}^{\text{auto}} \right| + \left| V_{\nu}^{\text{gd}} \right| \right)}, \text{ gJaccard} = \frac{\sum_{\nu=1}^{N_{\nu}} \left| V_{\nu}^{\text{auto}} \cap V_{\nu}^{\text{gd}} \right|}{\sum_{\nu=1}^{N_{\nu}} \left| V_{\nu}^{\text{auto}} \cup V_{\nu}^{\text{gd}} \right|}, \tag{12}$$

To evaluate the accuracy of boundary delineation, surface distance between the segmentation result and its gold standard can be used. Mean and standard deviation of the surface distance calculated from a set of discrete sample points of the surfaces are widely used to represent the distance between the two surfaces [11-15]. Table 3 and Table 4 present the validation results of the whole heart segmentation methods reported in the literature. The best accuracy reported were from Weese et al. [11-13] and Zheng et al. [14, 15]. The computation time of them were also efficient, as the former took 10 to 30 seconds on average to segment one volume, while the latter only needed 4 seconds, making it widely applicable and easily integrated into clinical practice. For the data used in the validation, Zheng et al. tested their algorithm on more than 500 CT volumes from 137 subjects [14, 15], and Zhuang et al. [19] validated their method on a set of clinical data involving nine different pathologies. The two tools demonstrated good robustness against clinical datasets with large shape variations, although the computation speed of the latter method needs to be improved (as long as 231 seconds [24]). It should be noted that due to the differences in test data sets and hardware, an objective comparison between these methods is difficult. Therefore, a grand challenge can be useful to achieve this goal in the future. Several useful challenges on cardiac segmentation, although not on the whole heart, have been organized, including the following:

- right ventricle segmentation: http://www.litislab.eu/rvsc
- atrium and the contrast enhanced region segmentation: http://www.isd.kcl.ac.uk/cdemris/
- coronary artery segmentation: http://coronary.bigr.nl/ http://grand-challenge2012.bigr.nl/
- left ventricle segmentation: http://smial.sri.utoronto.ca/LV_Challenge/ http://www.cardiacatlas.org/web/guest/stacom2011

5. DISCUSSIONS

The challenges of whole heart segmentation are mainly due to the large shape variations of the heart and the indistinct boundaries between substructures. To tackle these challenges, segmentation generally employs prior models to guide the procedure, which is formulated as a fitting process from the prior model to the unseen image. According to the different representations of the prior model, the existing segmentation frameworks can be classified into two groups: (1) the atlas propagation and segmentation methods, and (2) the deformable model using statistical shape model (SSM) approaches.

In the atlas propagation and segmentation methods, the atlas is relatively easy to construct and the construction only needs a small number of training data. This is because the atlas does not have built-in statistical information of the shape or appearance, and therefore the segmentation propagation mainly relies on the registration algorithms to deal with the challenges. Hierarchical registration approaches are adopted to improve the robustness of segmentation propagation. The shape regularization is implemented by combining a deformation smoothness term with a similarity term. Although researchers in the image computing field continue improving the registration algorithms, there is an increasing interest in using a large number of atlases for multiple atlas propagation and segmentation (MAPS). In MAPS, a small number of atlases, whose shapes (or both shapes and appearance) are most similar to that of the unseen image, are selected for segmentation propagation. Therefore, fast atlas-ranking algorithms play an important role in these segmentation frameworks. Recently, a new atlas-based segmentation method, the patch-based segmentation (PBS), was proposed for segmentation of brain MRI [110], and was also extended to segment multiple organs from abdominal CT scans [111]. PBS does not need to accurately register the selected atlases to the unseen images; for example, an affine registration can be used [110]. The segmentation propagation is achieved by ranking patches from the selected atlases and fusing label information of patches according to their rankings.

A practical problem of the atlas propagation and segmentation methods is the computation time. This is because a nonrigid registration process is generally needed in the atlas propagation, and this registration can be computationally expensive. In whole heart segmentation, one atlas propagation could take about 20 minutes [16, 17] or more than one hour [19]. The computation time of nonrigid registration represents a more significant issue in MAPS, although the computation time can be reduced to 231 seconds by simply re-implementing the registration algorithms [24]. PBS has similar computation complexity as MAPS, although nonrigid registration may not be needed [110, 111]. Recent advance in image registration can reduce the nonrigid registration to within a minute or even a few seconds when GPU programming is used [e.g., 112, 113]. Therefore, it is anticipated that these techniques will be applied to atlas propagation for whole heart segmentation.

The deformable model-based segmentation is normally computationally efficient. For example, the whole heart segmentation only takes 4 to 30 seconds [11–15]. This is because the fitting process is generally performed using local texture information of the unseen image. Such local search requires the segmentation approaches to closely initialize the surface elements of the prior model to the ground truth boundaries. SSM is widely adopted for the automatic segmentation of the model is constrained to be within the SSM-based framework, deformation of the model is constrained to be within the shape variability of the SSM, and thus the segmentation result is maintained with a realistic shape [22, 14, 15]. To capture the large shape variations of the heart, a large number of data, which represent all unseen images in clinics, are required for the

training of the SSM. This is however practically difficult. Several methods have been proposed in the literature to artificially enlarge the training set or relax the constraint from the prior shape model for automatic whole heart segmentation. Since the shapes of the segmentation results are not exactly guaranteed by the SSM, the fitting process should be elaborately designed to regularize the deformation of the surface model. For example, in the shape constrained deformable model framework, the hierarchical piecewise affine adaptation is used before the fully deformable adaptation [11–13]. Similar to the registration-based techniques, these approaches combine a shape regularization term and a similarity term within an energy function. The regularization term penalizes the deviation of the deformations to the prior model, and the similarity term determines the accuracy of the local adaptation of the model surface to the detected boundaries. The balance between them is controlled by the weighting parameters of the two terms.

Boundary detection is crucial in the deformable model-based segmentation, as the accuracy of the local adaptation is determined by the detected boundaries points. AAM builds the statistical information of the texture patterns of the objective organ, as well as the statistical information of the shapes. AAM can then be used in the intensity-based registration framework, where the parameters weighting both the shape eigen modes and the appearance eigen modes are optimized [28, 40, 73, 114]. These approaches can combine the advantages from both the atlas-based segmentation and the deformable model-based segmentation methods to improve the robustness and accuracy. However, besides the same practical problem of demanding a large number of training data, the dimension of the vectors representing both the texture and the shape of a whole heart image can be too large to apply the PCA. Therefore, methods which can compactly describe the texture patterns of the whole heart are desirable.

The AAM segmentation framework can be considered as a method to generate a subject-specific atlas for an unseen image using the learned image manifold. Therefore, other atlas generation methods can be considered in the future. Several works were proposed recently for learning a nonlinear manifold of face images [115], brain MRI [116, 117], and cardiac MRI [118]. The metadata, such as the subject age and sex, were included in the modeling of brain images, and the variables of the metadata controlled the training of the model [117]. Therefore, new instances can be easily generated for segmentation propagation, given that the specific values of these variables are available. In practice, the information of subject age and sex may not be enough to create a good atlas, and generating a subject-specific atlas can be computationally demanding using regression methods of Gerber et al. [116] and Rohlfing et al. [117]. Fast computation on a shape manifold was proposed to generate subject-specific atlases [118]. The new atlas of an unseen image is formulated as the Frechet mean of existing atlases, which can be directly computed via fast vector operations on velocity fields. Sparse shape composition is another modeling technique which creates new instances by linearly combining a small number of existing samples [100, 101]. The method has been shown to be efficient for shape modeling and automated segmentation of multiple organs. In the future, the method is expected to include appearance of images and to be extended to the segmentation of the whole heart.

It should be noted that most of the prior model-based segmentation methods can only deal with the images which have the same topology as the prior model. The difference of topology represents a more challenging task than the shape variations from different pathologies. Different pathologies may induce very different heart shapes, but the topology of the heart may still be the same. No work has been reported to study the segmentation of images with different topologies, and the existing approaches reviewed in this article may fail in such situation. For instance, it is difficult to segment an unseen volume from a patient with dextro-transposition of the great arteries using a prior model constructed from cardiac images with normal topology of the heart. In such patients, the aorta and the pulmonary artery are malformationally connected to the right ventricle and the left ventricle, respectively. To deal with such cases, a potential solution is to construct an atlas (or deformable model) *pool* which has at least one instance corresponding to one anatomical topology of the heart, and design an efficient atlas ranking-and-selection algorithm to automatically select the corresponding atlases for segmentation. In practice, the metadata, such as pathologies and stages of the pathology, can be useful in the atlas selection algorithm.

6. CONCLUSIONS

This work reviews the existing techniques of fully automatic whole heart segmentation, and analyzes the challenges and methodologies of the computation task. The challenges are in large variability of the heart shape, indistinct boundaries, and limited image quality. The reported methods generally employed prior models and formulated the segmentation as a fitting process from the prior model to the unseen images.

To analyze the methodologies, the existing works are classified in terms of the prior models and the techniques used in the fitting process. The prior models include the atlas (single atlas or multiple atlases) and the SSM-based deformable models. In general, the fitting process is implemented in a hierarchy fashion, which is further decomposed into three stages: localization of the whole heart, initialization of substructures, and refinement of boundary delineation. Also, to guarantee a segmentation result with a realistic shape, shape regularization is needed in the deformable fitting. Table 1 classifies whole heart segmentation works in terms of the types of techniques, Table 2 lists the techniques for the works, and Table 3 and Table 4 summarize the reported performance of each method, although an objective comparison is not possible and an open-access dataset with gold standard is in need.

Future research is needed to build robust, fast and computationally inexpensive whole heart segmentation methods for clinical applications with a wide range of pathologies. In the model-based segmentation framework, to ease the fitting process, the prior model is expected to resemble the unseen image by simple transformations. Therefore, all available priors, ideally with information of statistical distributions, should be built into the model. For example, a large amount of data are used for the multiple atlases-based segmentation or for the training of statistical shape models. To reduce the complexity of model construction, it is necessary to design fitting algorithms that are capable of registering the model to an unseen image even when they are very different. Finally, extending the applicability of the existing priors is an important technology, especially when the available training data are limited, such as generating new atlases using existing ones and extending flexibility of the SSM.

ACKNOWLEDGEMENTS

The author would like to thank the reviewers and editors for their constructive comments that enormously improved the presentation of this paper.

CONFLICT OF INTEREST

The author indicated no potential conflicts of interest.

REFERENCES

- [1] World Health Organization: Cardiovascular diseases (CVDs) Fact Sheet No. 317, http://www.who.int/mediacentre/factsheets/fs317/en/index.html. Accessed November 1, 2012.
- [2] Van der Geest RJ, Reiber JHC. Quantification in Cardiac MRI. Journal of Magnetic Resonance Imaging, 1999, 10: 602–608.
- [3] Frangi AF, Niessen WJ, Viergever MA. Three-dimensional Modeling for functional analysis of cardiac images. A review. *IEEE Transactions on Medical Imaging*, 2001, 20, 2–5.
- [4] Takuma S, Ota T, Muro T, Hozumi T, Sciacca R, Di Tullio MR, Blood DK, Yoshikawa J, Homma S. Assessment of left ventricular function by real-time 3-dimensional echocardiography compared with conventional noninvasive methods. *Journal of the American Society of Echocardiography*, 2001, 14:275–284.
- [5] Nesser HJ, Sugeng L, Corsi C, Weinert L, Niel J, Ebner C, Steringer-Mascherbauer R, Schmidt F, Schummers G, Lang RM, Mor-Avi V. Volumetric analysis of regional left ventricular function with real-time three-dimensional echocardiography: validation by magnetic resonance and clinical utility testing. *Heart*, 2007: 572–578.
- [6] Chartrand-Lefebvre C, Cadrin-Chênevert A, Bordeleau E, Ugolini P, Ouellet R, Sablayrolles JL, Prenovault J. Coronary computed tomography angiography: overview of technical aspects, current concepts, and perspectives. *Canadian Association of Radiologists Journal*, 2007, 58: 92–108.
- [7] Earls JP, Ho VB, Foo TK, Castillo E, Flamm SD. Cardiac MRI: Recent Progress and continued challenges. *Journal of Magnetic Resonance Imaging*, 2002, 16: 111–127.
- [8] Ecabert O, Peters J, Walker MJ, Ivanc T, Lorenz C, Berg J, Lessick J, Vembar M, Weese J. Segmentation of the heart and great vessels in CT images using a model-based adaptation framework. *Medical Image Analysis*, 2011, 15: 863–876.
- [9] Koikkalainen J, Tolli T, Lauerma K, Antila K, Mattila E, Lilja M, Lotjonen J, Tölli T, Lötjönen J. Methods of Artificial Enlargement of the Training Set for Statistical Shape Models. *IEEE Transaction* on Medical Imaging, 2008, 27: 1643–1654.
- [10] Ecabert O, Peters J, Lorenz C, Von Berg J, Vembar M, Subramanyan K, Lavi G, Weese J. Towards automatic full heart segmentation in computed-tomography images. *Computers in Cardiology*, 2005, 223–226.
- [11] Ecabert O, Peters J, Schramm H, Lorenz C, Von Berg J, Walker MJ, Vembar M, Olszewski ME, Subramanyan K, Lavi G, Weese J. Automatic Model-Based Segmentation of the Heart in CT Images. *IEEE transactions on medical imaging*, 2008, 27: 1189–201.
- [12] Peters J, Ecabert O, Meyer C, Schramm H, Kneser R, Groth A, Weese J. Automatic Whole Heart Segmentation in Static Magnetic Resonance Image Volumes. *Medical Image Computing and Computer Assisted Intervention*, 2007: 402–410.
- [13] Peters J, Ecabert O, Meyer C, Kneser R, Weese J. Optimizing boundary detection via Simulated Search with applications to multi-modal heart segmentation. *Medical Image Analysis*, 2009, 14: 70–84.

- [14] Zheng Y, Georgescu B, Barbu A, Scheuering M, Comaniciu D. Four-chamber heart modeling and automatic segmentation for 3D cardiac CT volumes. *Proceedings of SPIE*, 2008, 6914: 691416–691416–12.
- [15] Zheng Y, Barbu A, Georgescu B, Scheuering M, Comaniciu D. Four-chamber heart modeling and automatic segmentation for 3-D cardiac CT volumes using marginal space learning and steerable features. *IEEE transactions on medical imaging*, 2008, 27: 1668–81.
- [16] Kirisli HA, Schaap M, Klein S, Neefjes L, Wustink A, Van Walsum T, Niessen J. Fully automatic cardiac segmentation from 3D CTA data: a multi-atlas based approach, *Proc. of SPIE, Medical Imaging: Image Processing*, 2010: 1–9.
- [17] Kirisli HA, Schaap M, Klein S, Papadopoulou SLL, Bonardi M, Chen CHH, Weustink ACC, Mollet NRR, Vonken EJJ, Van der Geest RJJ, Van Walsum T, Niessen WJJ. Evaluation of a multi-atlas based method for segmentation of cardiac CTA data: a large-scale, multicenter, and multivendor study. *Medical Physics*, 2010, 37: 6279–6291.
- [18] Zhuang X, Rhode K, Arridge S, Razavi R, Hill D, Hawkes D, Ourselin S. An Atlas-Based Segmentation Propagation Framework Using Locally Affine Registration –– Application to Automatic Whole Heart Segmentation. In: Metaxas, D, Axel, L, Fichtinger, G, and Székely, G. (eds) *Medical Image Computing and Computer Assisted Intervention*, 2008: 425–433.
- [19] Zhuang X, Rhode K, Razavi R, Hawkes DJ, Ourselin S. A Registration-Based Propagation Framework for Automatic Whole Heart Segmentation of Cardiac MRI. *IEEE transactions on medical imaging*, 2010, 29, 1612–1625.
- [20] Zhuang X, Leung K, Rhode K, Razavi R, Hawkes DJ, Ourselin S. Whole Heart Segmentation of Cardiac MRI Using Multiple Path Propagation Strategy. *Medical Image Computing and Computer Assisted Intervention*, 2010: 435–443.
- [21] Murphy S, Akinyemi A, Steel J, Petillot Y, Poole I. Multi-compartment heart segmentation in CT angiography using a spatially varying gaussian classifier. *International journal of computer assisted* radiology and surgery, 2012.
- [22] Lotjonen J, Kivisto S, Koikkalainen J, Smutek D, Lauerma K, Lötjönen J, Kivistö S. Statistical shape model of atria, ventricles and epicardium from short- and long-axis MR images. *Medical Image Analysis*, 2004, 8: 371–386.
- [23] Zhuang X. <u>Automatic whole heart segmentation based on image registration</u>. PhD dissertation, University College London, 2010.
- [24] Zhuang X, Song J, Zhan S, Lan T, Huang H, Hu M, Ourselin S, Li Q. A registration and atlas propagation based framework for automatic whole heart segmentation of CT volumes. *Proc. SPIE* 8669, *Medical Imaging: Image Processing*, 2013: 86693W.
- [25] Kaus MR, Von Berg J, Weese J, Niessen WJ, Pekar V. Automated segmentation of the left ventricle in cardiac MRI. *Medical Image Analysis*, 2004, 8(3):245–54.
- [26] Jolly MP. Automatic Segmentation of the Left Ventricle in Cardiac MR and CT Images. International Journal of Computer Vision, 2006, 70: 151–163.
- [27] Lorenzo-Valdes M, Sanchez-Ortiz GI, Mohiaddin R, Rueckert D. Segmentation of 4D Cardiac MR Images Using a Probabilistic Atlas and the EM Algorithm. In: Ellis, R.E. and Peters, T.M. (eds) *Medical Image Computing and Computer Assisted Intervention*, 2003, Part I: 440–450.
- [28] Lapp RM, Lorenzo-Valdes M, Rueckert D. 3D/4D cardiac segmentation using active appearance models, non-rigid registration and the Insight Toolkit. *Medical Image Computing and Computer Assisted Intervention*, 2004: 419–426.
- [29] Lorenzo-Valdes M, Sanchez-Ortiz GI, Mohiaddin R, Rueckert D. Atlas-based Segmentation and Tracking of 3D cardiac MR images using Non-rigid Registration. In: Dohi, T. and Kikinis, R. (eds) Medical Image Computing and Computer Assisted Intervention, 2002: 642–650.
- [30] Lorenzo-Valdes M, Sanchez-Ortiz GI, Elkington AG, Mohiaddin R, Rueckert D. Segmentation of 4D cardiac MR images using a probabilistic atlas and the EM algorithm. *Medical Image Analysis*, 2004, 8: 255–265.

- [31] Montagnat J, Delingette H. 4D deformable models with temporal constraints: Application to 4D cardiac image segmentation. *Medical Image Analysis*, 2005, 9: 87–100.
- [32] Pluempitiwiriyawej C, Moura JMF, Wu YJL, Ho C. STACS: New active contour scheme for cardiac MR image segmentation. *IEEE transactions on medical imaging*, 2005, 24: 593–603.
- [33] Kurkure U, Pednekar A, Muthupillai R, Flamm SD, Kakadiaris IA. Localization and segmentation of left ventricle in cardiac cine-MR images. *IEEE Transactions on Bio-Medical Engineering*, 2009, 56: 1360–1370.
- [34] Van Assen HC, Danilouchkine MG, Frangi AF, Ords S, Westenberg JJM, Reiber JHC, Lelieveldt BPF. SPASM: A 3D-ASM for segmentation of sparse and arbitrarily oriented cardiac MRI data. *Medical Image Analysis*, 2006, 10: 286–303.
- [35] Van Assen HC, Danilouchkine MG, Dirksen MS, Reiber JHC, Lelieveldt BPF. A 3-D Active Shape Model Driven by Fuzzy Inference: Application to Cardiac CT and MR. *IEEE Transactions on Information Technology in Biomedicine*, 2008, 12: 595–605.
- [36] Rajpoot K, Noble JA, Grau V, Szmigielski C, Becher H. Image-Driven Cardiac Left Ventricle Segmentation for the Evaluation of Multiview Fused Real-Time Dimensional Echocardiography Images. *Medical Image Computing and Computer Assisted Intervention*, 2009: 893–900.
- [37] Rajpoot K, Noble JA, Grau V, Szmigielski C, Becher H. Multiview RT3D Echocardiography Image Fusion. In: Ayache, N, Delingette, H, and Sermesant, M. (eds). *Functional Imaging and Modeling of* the Heart, 2009: 134–143.
- [38] Shi W, Zhuang X, Wang H, Duckett S, Oregan D, Edwards P, Ourselin S, Rueckert D. Automatic segmentation of different pathologies from cardiac cine MRI using registration and multiple component EM estimation. *Functional Imaging and Modeling of the Heart*, 2011: 163–170.
- [39] Fritscher KD, Pilgram R, Schubert R. Automatic Cardiac 4D Segmentation Using Level Sets. In: Frangi, A.F, Radeva, P, Santos, A, and Hernandez, M. (eds). *Functional Imaging and Modeling of the Heart*, 2005:113–122.
- [40] Mitchell SC, Bosch JG, Lelieveldt BPF, Van der Geest RJ, Reiber JHC, Sonka M. Active Appearance Models: Segmentation of Cardiac MR and Ultra sound Images. *IEEE Transactions on Medical Imaging*, 2002, 21: 1167–1178.
- [41] Ben Ayed I, Chen HM, Punithakumar K, Ross I, Li S. Max-flow segmentation of the left ventricle by recovering subject-specific distributions via a bound of the Bhattacharyya measure. *Medical Image Analysis*, 2012, 16: 87–100.
- [42] Cocosco CA, Niessen WJ, Netsch T, Vonken A, Lund G, Stork A, Viergever A. Automatic imagedriven segmentation of the ventricles in cardiac cine MRI. *Journal of Magnetic Resonance Imaging*, 2008, 28: 366–374.
- [43] Andreopoulos A, Tsotsos JK. Efficient and generalizable statistical models of shape and appearance for analysis of cardiac MRI. *Medical Image Analysis*, 2008, 12: 335–357.
- [44] Casero R, Burton R, Quinn T, Bollensdorff C, Hales P, Schneider JE, Kohl P, Grau V. Cardiac Valve Annulus Manual Segmentation Using Computer Assisted Visual Feedback in Three-Dimensional Image Data. *IEEE Eng Med Biol Soc.*, 2010: 738–41.
- [45] Boykov Y, Florin C, Jolly M, Ramaraj R, Rinck D. AUTOMATIC HEART ISOLATION FOR CT CORONARY VISUALIZATION USING GRAPH-CUTS. *ISBI*, 2006: 614–617.
- [46] Isgum I, Staring M, Rutten A, Prokop M, Viergever MA, Van Ginneken B. Multi-atlas-based segmentation with local decision fusion—application to cardiac and aortic segmentation in CT scans. *IEEE transactions on medical imaging*, 2009, 28: 1000–10.
- [47] Kang D, Woo J, Slomka PJ, Dey D, Germano G, Jay Kuo CC. Heart chambers and whole heart segmentation techniques: review. *Journal of Electronic Imaging*, 2012, 21: 010901.
- [48] Ordas S, Frangi AF. Automatic Quantitative Analysis of Myocardial Wall Motion and Thickening from Long-and Short-Axis Cine MRI Studies. *IEEE Engineering in Medicine and Biology Society*, 2005: 7028–7031.

- [49] Frangi A, Rueckert D, Duncan JS. Three-dimensional cardiovascular image analysis. IEEE transactions on medical imaging, 2002, 21: 1005–10.
- [50] Shkhvatsabaia I. Intracardiac hemodynamics and clinico-pathogenetic variants of the course of hypertension. *Kardiologiia*, 1977, 17: 8–18.
- [51] Juenger J, Schellberg D, Kraemer S, Haunstetter A, Zugck C, Herzog W, Haass M. Health related quality of life in patients with congestive heart failure: comparison with other chronic diseases and relation to functional variables. *Heart*, 2002, 87: 235–241.
- [52] Owan TE, Hodge DO, Herges RM, Jacobsen SJ, Roger VL, Redfield MM. Trends in Prevalence and Outcome of Heart Failure with Preserved Ejection Fraction. *The New England Journal of Medicine*, 2006, 355: 251–259.
- [53] Shi W, Zhuang X, Wang H, Duckett S, Luong N, Tobon-gomez C, Tung K, Edwards PJ, Rhode KS, Razavi RS, Ourselin S, Rueckert D. A comprehensive cardiac motion estimation framework using both untagged and 3D tagged MR images based on non-rigid registration. *IEEE transactions on medical imaging*, 2012, 31: 1263–1275.
- [54] Duckett SG, Shi W, Zhuang X, Shetty A, Ginks M, Rinaldi C, Carr-White G, Rueckert D, Razavi RS. Cardiac MRI: understanding myocardial motion to predict remodelling pre cardiac resynchronisation therapy. *Heart*, 2012, 98: A6–A7.
- [55] Dougherty AH, Naccarelli GV, Gray EL, Hicks CH, Goldstein RA. Congestive heart failure with normal systolic function. *The American Journal of Cardiology*, 1984, 54: 778–782.
- [56] Vasan RS, Benjamin EJ, Levy D. Congestive Heart Failure With Normal Left Ventricular Systolic Function Clinical Approaches to the Diagnosis and Treatment of Diastolic Heart Failure. *Arch Intern Med.*, 1996, 156: 146–157.
- [57] Fletcher BD, Jacobstein MD, Nelson AD, And TAR, Alfidi RJ. Gated magnetic resonance imaging of congenital cardiac malformations. *Radiology*, 1984, 150: 137–140.
- [58] Shekhar R, Zagrodsky V. Cine MPR: interactive multiplanar reformatting of four-dimensional cardiac data using hardware-accelerated texture mapping. *IEEE Trans Inf Technol Biomed*, 2003, 7: 384–93.
- [59] Ma Y, Karim R, Housden R. Cardiac Unfold: A Novel Technique for Image-Guided Cardiac Catheterization Procedures. *Information Processing in Computer-Assisted Interventions*, 2012: 104–114.
- [60] Pal NR, Pal SK. A Review on Image Segmentation Techniques. Pattern Recognition, 1993, 26: 1277–1294.
- [61] Bezdek JC, Hall LO, Clarke LP. Review of MR image segmentation techniques using pattern recognition. *Medical Physics*, 1993, 20: 1033–1048.
- [62] Pham DL, Xu C, Prince JL. Current Methods in Medical Image Segmentation. Annual Review of Biomedical Engineering, 2000, 02: 315–337
- [63] Suri JS. Computer Vision, Pattern Recognition and Image Processing in Left Ventricle Segmentation: The Last 50 Years. *Pattern Analysis and Applications*, 2000, 3: 209–242
- [64] Zhang H, Fritts JE, Goldman S. Image segmentation evaluation: A survey of unsupervised methods. *Computer Vision and Image Understanding*, 2008, 110: 260–280.
- [65] Cremers D, Rousson M, Deriche R. A review of statistical approaches to level set segmentation: integrating color, texture, motion and shape. *International journal of computer vision*, 2007, 72: 195–215.
- [66] Petitjean C, Dacher JN. A review of segmentation methods in short axis cardiac MR images. *Medical Image Analysis*, 2011, 15: 169–84.
- [67] Rueckert D, Lorenzo-Valdeés M, Chandrashekara R, Sanchez-Ortiz GI, Mohiaddin R. Non-rigid registration of cardiac MR: Application to motion modelling and atlas-based segmentation. *IEEE International Symposium on Biomedical Imaging*, 2002: 481–484.
- [68] Weese J, Kaus M, Lorenz C, Lobregt S, Truyen R, Pekar V. Shape Constrained Deformable Models for 3D Medical Image Segmentation. In: Insana, M.F. and Leahy, R.M. (eds). *Information Processing in Medical Imaging*, 2001: 380–387.

- [69] Rohlfing T, Maurer JCR. Multi-classifier framework for atlas-based image segmentation. Pattern Recognition Letters, 2005, 26: 2070–2079.
- [70] Van Rikxoort EM, Isgum I, Arzhaeva Y, Staring M, Klein S, Viergever MA, Pluim JPW, Van Ginneken, B. Adaptive local multi-atlas segmentation application to the heart and the caudate nucleus. *Medical Image Analysis*, 2010, 14(1): 39–49.
- [71] Cootes TF, Taylor CJ, Cooper DH, Graham J. Active Shape Models: Their Training and Application. Computer Vision and Image Understanding, 1995, 61: 38–59.
- [72] Heimann T, Meinzer HP. Statistical shape models for 3D medical image segmentation: a review. *Medical Image Analysis*, 2009, 13: 543–63.
- [73] Cootes TF, Edwards GJ, Taylor CJ. Active appearance models. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 2001, 23: 681–685.
- [74] Hautvast G, Lobregt S, Breeuwer M, Gerritsen F. Automatic Contour Propagation in Cine Cardiac Magnetic Resonance Images. *IEEE transactions on medical imaging*, 2006, 25: 1472–1482.
- [75] Rouchdy Y, Pousin JJ, Schaerer JJ, Clarysse P. A nonlinear elastic deformable template for soft structure segmentation: Application to the heart segmentation in MRI. *Inverse Problems*, 2007, 23: 1017–1035.
- [76] Dawant BM, Hartmann SL, Thirion JP, Maes F, Vandermeulen D, Demaerel P. Automatic 3-D Segmentation of Internal Structures of the Head in MR Images Using a Combination of Similarity and Free-Form Transformations: Part I, Methodology and Validation on Normal Subjects. *IEEE transactions on medical imaging*, 1999, 18: 909–916.
- [77] Heckemann RA, Hajnal JV, Aljabar P, Rueckert D, Hammers, A. Automatic anatomical brain MRI segmentation combining label propagation and decision fusion. *NeuroImage*, 2006, 33: 115–126.
- [78] Aljabar P, Heckemann RA, Hammers A, Hajnal JV, Rueckert D. Multi-atlas based segmentation of brain images: Atlas selection and its effect on accuracy. *NeuroImage*, 2009, 46: 726–738.
- [79] Lotjonen JMP, Wolz R, Koikkalainen JR, Thurfjell L, Waldemar G, Soininen H, Rueckert D. Fast and robust multi-atlas segmentation of brain magnetic resonance images. *Neuroimage*, 2010, 49: 2352–2365.
- [80] Maes F, Collignon A, Vandermeulen D, Marchal G, Suetens P. Multimodality image registration by Maximization of Mutual information. *IEEE Transactions on Medical Imaging*, 1997, 16: 187–198.
- [81] Viola P, Wells III VM. Alignment by Maximization of Mutual Information. International Journal of Computer Vision, 1997, 24: 137–154.
- [82] Studholme C, Hill DLG, Hawkes DJ. An overlap invariant entropy measure of 3D medical image alignment. *Pattern Recognition*, 1999, 32: 71–86.
- [83] Kittler J, Hatef M, Duin RPW, Matas J. On combining classifiers. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 1998, 20: 226–239.
- [84] Kass M, Witkin A, Terzopoulos D. Snakes: Active Contour Models. International Journal of Computer Vision, 1988, 1: 321–331.
- [85] McInerney T, Terzopoulos D. Deformable models in Medical Image Analysis: a survey. Medical Image Analysis, 1996, 1: 91–108.
- [86] Lotjonent J, Antila K, Koikkalainen J, Lilja M, Cootes T. Artificial Enlargement of a Training Set for Statistical Shape Models?: Application to Cardiac Images. *Functional Imaging and Modeling of the Heart*, 2005: 92–101.
- [87] Staib L, Wang Y. Boundary finding with prior shape and smoothness models. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 2000, 22: 738–743.
- [88] Cootes T, Taylor C. Combining point distribution models with shape models based on finite element analysis. *Image and Vision Computing*, 1995, 13: 403–409.
- [89] Davatzikos C, Tao X, Shen D. Hierarchical active shape models, using the wavelet transform. *IEEE transactions on medical imaging*, 2003, 22: 414–23.
- [90] Zhao Z, Aylward SR, Teoh E. A novel 3D partitioned active shape model for segmentation of brain MR images. *Medical Image Computing and Computer Assisted Intervention*, 2005, 8: 221–8.

- [91] Davatzikos C, Shen D, Mohamed A. A framework for predictive modeling of anatomical deformations. *IEEE transactions on medical imaging*, 2001, 20(8): 836–843.
- [92] Cootes T, Hill A, Taylor CJ, Haslam J, Manchester M. The Use of Active Shape Models for Locating Structures in *Medical Images. Image and Vision Computing*, 1994, 12: 355–366.
- [93] Shen D, Herskovits E, Davatzikos C. Adaptive-focus statistical shape model for segmentation and shape modeling of 3-D brain structures. *IEEE Transactions on Medical Imaging*, 2001, 20: 257–270.
- [94] Mohamed A, Davatzikos C, Taylor R. A combined statistical and biomechanical model for estimation of intra-operative prostate deformation. *Medical Image Computing and Computer Assisted Intervention*, 2002: 452–460.
- [95] Hu Y, Ahmed HU, Taylor Z, Allen C, Emberton M, Hawkes D, Barratt D. MR to ultrasound registration for image-guided prostate interventions. *Medical Image Analysis*. 16, 687–703 (2012).
- [96] Siddiqi K, Pizer S. <u>Medial Representations: Mathematics, Algorithms and Applications</u>. Springer Verlag, 2008.
- [97] Pizer SM, Fletcher PT, Joshi SC, Thall A, Chen JZ, Fridman Y, Fritsch DS, Lu A, Muller KE, Tracton G, Yushkevich PA, Chaney EL. Deformable M-Reps for 3D Medical Image Segmentation. *International Journal of Computer Vision*, 2003, 55: 85–106.
- [98] Pilgram R, Schubert R, Fritscher KD, Zwick RH, Schocke MF, Trieb T, Pachinger O. Shape Discrimination of Healthy and Diseased Cardiac Ventricles using Medial Representation. *International Journal of Computer Assisted Radiology and Surgery*, 2006, 1: 33–38.
- [99] Sermesant M, Moireau P, Camara O, Sainte-Marie J, Andriantsimiavona R, Cimrman R, Hill D, Chapelle D, Razavi R. Cardiac function estimation from MRI using a heart model and data assimilation: advances and difficulties. *Medical Image Analysis*, 2006, 10: 642–56.
- [100] Zhang S, Zhan Y, Dewan M, Huang J, Metaxas DN, Zhou XS. Deformable Segmentation via Sparse Shape Representation. *Medical Image Computing and Computer Assisted Intervention*, 2011: 451–458.
- [101] Zhang S, Zhan Y, Metaxas DN. Deformable Segmentation via Sparse Representation and Dictionary Learning. *Medical Image Analysis*, 2012, 16: 1385–1396.
- [102] Zhang D, Lu G. Review of shape representation and description techniques. *Pattern Recognition*, 2004, 37: 1–19.
- [103] Gao M, Huang J, Zhang S, Qian Z, Voros S, Metaxas D, Axel L. 4D cardiac reconstruction using high resolution CT images. *Functional Imaging and Modeling of the Heart*, 2011: 153–160.
- [104] Pluim J, Maintz J, Viergever M. Mutual Information Based Registration of Medical Images: A Survey. IEEE Transactions on Medical Imaging, 2003, 22: 986–1004.
- [105] Rueckert D, Sonoda LI, Hayes C, Hill DLG, Leach MO, Hawkes DJ. Nonrigid registration using freeform deformations: Application to breast MR images. *IEEE transactions on medical imaging*, 1999, 18: 712–721.
- [106] Crum WR, Tanner C, Hawkes DJ. Anisotropic multi-scale fluid registration: Evaluation in magnetic resonance breast imaging. *Physics in Medicine and Biology*, 2005, 50: 5153–5174.
- [107] Dice LR. Measures of the amount of ecologic association between species. *Ecology*, 1945, 26: 297–302.
- [108] Jaccard P. Étude comparative de la distribution florale dans une portion des Alpes et des Jura. Bulletin de la Société Vaudoise des Sciences Naturelles, 1901, 37: 547–579.
- [109] Crum W, Camara O, Hill D. Generalized overlap measures for evaluation and validation in Medical Image Analysis. *IEEE Transactions on Medical Imaging*, 2006, 25: 1451–61.
- [110] Coupé P, Manjón JV, Fonov V, Pruessner J, Robles M, Collins DL. Patch-based segmentation using expert priors: application to hippocampus and ventricle segmentation. *NeuroImage*, 2011, 54: 940–54.
- [111] Wolz R, Chu C, Misawa K. Multi-organ Abdominal CT Segmentation Using Hierarchically Weighted Subject-Specific Atlases. *Medical Image Computing and Computer Assisted Intervention*, 2012: 10–17.

- [112] Glocker B, Komodakis N, Tziritas G, Navab N, Paragios N. Dense image registration through MRFs and efficient linear programming. *Medical Image Analysis*, 2008,12: 731–41.
- [113] Modat M, Ridgway G, Taylor Z, Lehmann M, Barnes J, Hawkes DJ, Fox NC, Ourselin S. Fast freeform deformation using graphics processing units. *Computer Methods and Programs in Biomedicine*, 2010, 98(3):278-84.
- [114] Cootes TF, Beeston C, Edwards GJ, Taylor CJ. A Unified Framework for Atlas Matching Using Active Appearance Models. *Lecture Notes in Computer Science*, 1999, 1613: 322–333.
- [115] Lin T, Zha H. Riemannian manifold learning. *IEEE Pattern Analysis and Machine Intelligence*, 2008, 30: 796–809.
- [116] Gerber S, Tasdizen T, Fletcher PT, Joshi S, Whitaker R. Manifold modeling for brain population analysis. *Medical Image Analysis*, 2010, 14: 643–653.
- [117] Rohlfing T, Sullivan EV, Pfefferbaum A, Pfefferbaum A. Regression models of atlas appearance. *Information processing in medical imaging*, 2009, 21: 151–62.
- [118] Zhuang X, Shi W, Wang H, Rueckert D, Ourselin S. Computation on shape manifold for atlas generation: application to whole heart segmentation of cardiac MRI. *Proc. SPIE, Medical Imaging: Image Processing*, 2013, 8669: 866941.





The Scientific

World Journal

Machinery

Rotating

Journal of Sensors



International Journal of Distributed Sensor Networks



Advances in Civil Engineering





Submit your manuscripts at http://www.hindawi.com









International Journal of Chemical Engineering





International Journal of Antennas and Propagation





Active and Passive Electronic Components





Shock and Vibration





Advances in Acoustics and Vibration